Author's response to reviews

Title: Targeting glycolysis by 3-Bromopyruvate improves Tamoxifen Cytotoxicity of Breast Cancer Cell Lines.

Authors:

Yasmin M Attia 1 (yasmin.m.attia@gmail.com)
Hanan S EL-Abhar 2 (hanan.elabhar@pharma.cu.edu.eg)
Mahmoud M Al Marzabani 1 (mahmoud.almarzabani@nci.cu.edu.eg)
Samia A Shouman 1 (samia.shouman@nci.cu.edu.eg)

Version: 6 Date: 17 October 2015

Author's response to reviews: see over
October, 12, 2015

Dear Editor:

We do appreciate and thank you very much for your valuable comments and suggestions that will improve the quality of our manuscript.

Editor comments

1-Reporting guidelines
In accordance with BioMed Central editorial policies, could you please ensure your manuscript adheres to ARRIVE guidelines (http://www.nc3rs.org.uk/arrive-guidelines) for reporting research using animals. This is to maximise the information published and minimise animal suffering. Can you please include a completed ARRIVE checklist as an additional file when submitting your revised manuscript.

Authors’ response

1-In agreement with journal policies, we checked our manuscript to adhere to ARRIVE guidelines for reporting research using animals, please find the attached files of the ARRIVE checklist and the corrected manuscript.

Editor comments

2. Table
Please label the table that starts on page 5 as "Table 1" and move it, adding a table legend, to the end of the manuscript. Please rephrase the sentence that is currently preceding it so that it makes a reference to table 1.

Authors’ response

2-According to the instruction of the editor we add label to table on page 5 and move it at the end of the manuscript. We add sentence so that it makes a reference to table 1.

Editor comments

3. Figures
we note that both Figures 6 and 7 span over several pages. Please note that each figure should fit on a single page in portrait format. If you are unable to submit Figures 6 and 7 as a composite images that fit on one single page each, you may wish to consider dividing the images into separate figures (e.g. Figure 6c-d would become a new Figure 7, Figure 6e-f and would become a new Figure 8, and so forth). If you chose to do so, please ensure that the corresponding changes are made when referring to figures in the manuscript.
Authors’ response

3-We divided the images into separate figures (Figure 7, new Figure 8) Sorry for the incorrect format.

Editor comments

4. Authors' contributions
Please use the author initials instead of the first names (e.g. YA instead of Yasmin), and please remove the superfluous "who" in the beginning of this section.

Authors’ response

4-We correct the names in Authors' contributions.

Editor comments

5. References
please correct the publication year of reference 22.

Authors’ response

5-Thank you we correct the reference.

Thank you